Georgia State University ScholarWorks @ Georgia State University

Public Health Faculty Publications

School of Public Health

2015

Impact of School Cycles and Environmental Forcing on the Timing of Pandemic Influenza Activity in Mexican States, May-December 2009

James Tamerius University of Iowa, james-tamerius@uiowa.edu

Cecile Viboud National Institutes of Health, Cecile.Viboud2@nih.gov

Jeffrey Shaman Columbia University, jls106@columbia.edu

Gerardo Chowell Georgia State University, gchowell@gsu.edu

Follow this and additional works at: https://scholarworks.gsu.edu/iph_facpub

Recommended Citation

Tamerius J, Viboud C, Shaman J, Chowell G (2015) Impact of School Cycles and Environmental Forcing on the Timing of Pandemic Influenza Activity in Mexican States, May-December 2009. PLoS Comput Biol 11(8): e1004337. doi:10.1371/journal.pcbi.1004337

This Article is brought to you for free and open access by the School of Public Health at ScholarWorks @ Georgia State University. It has been accepted for inclusion in Public Health Faculty Publications by an authorized administrator of ScholarWorks @ Georgia State University. For more information, please contact scholarworks@gsu.edu.

RESEARCH ARTICLE

Impact of School Cycles and Environmental Forcing on the Timing of Pandemic Influenza Activity in Mexican States, May-December 2009

James Tamerius¹*, Cécile Viboud², Jeffrey Shaman³, Gerardo Chowell⁴

1 Department of Geographical and Sustainability Sciences, University of Iowa, Iowa City, Iowa, United States of America, 2 Fogarty International Center, National Institutes of Health, Bethesda, Maryland, United States of America, 3 Environmental Health Sciences, Columbia University, New York, New York, United States of America, 4 School of Public Health, Georgia State University, Atlanta, Georgia, United States of America

* james-tamerius@uiowa.edu

Abstract

While a relationship between environmental forcing and influenza transmission has been established in inter-pandemic seasons, the drivers of pandemic influenza remain debated. In particular, school effects may predominate in pandemic seasons marked by an atypical concentration of cases among children. For the 2009 A/H1N1 pandemic, Mexico is a particularly interesting case study due to its broad geographic extent encompassing temperate and tropical regions, well-documented regional variation in the occurrence of pandemic outbreaks, and coincidence of several school breaks during the pandemic period. Here we fit a series of transmission models to daily laboratory-confirmed influenza data in 32 Mexican states using MCMC approaches, considering a meta-population framework or the absence of spatial coupling between states. We use these models to explore the effect of environmental, school-related and travel factors on the generation of spatially-heterogeneous pandemic waves. We find that the spatial structure of the pandemic is best understood by the interplay between regional differences in specific humidity (explaining the occurrence of pandemic activity towards the end of the school term in late May-June 2009 in more humid southeastern states), school vacations (preventing influenza transmission during July-August in all states), and regional differences in residual susceptibility (resulting in large outbreaks in early fall 2009 in central and northern Mexico that had yet to experience fullydeveloped outbreaks). Our results are in line with the concept that very high levels of specific humidity, as present during summer in southeastern Mexico, favor influenza transmission, and that school cycles are a strong determinant of pandemic wave timing.



Citation: Tamerius J, Viboud C, Shaman J, Chowell G (2015) Impact of School Cycles and Environmental Forcing on the Timing of Pandemic Influenza Activity in Mexican States, May-December 2009. PLoS Comput Biol 11(8): e1004337. doi:10.1371/journal. pcbi.1004337

Editor: Katia Koelle, Duke University, UNITED STATES

Received: November 24, 2014

Accepted: May 8, 2015

Published: August 20, 2015

Copyright: This is an open access article, free of all copyright, and may be freely reproduced, distributed, transmitted, modified, built upon, or otherwise used by anyone for any lawful purpose. The work is made available under the <u>Creative Commons CC0</u> public domain dedication.

Data Availability Statement: These data cannot be shared publicly due to privacy concerns. However, the data are available from the Mexican Social Security Institute (IMSS) for researchers who meet the criteria for access to confidential data. Data can be requested from the following website: http://www.imss.gob.mx/contacto.

Funding: Funding was provided by the NIH Models of Infectious Disease Agent Study program through cooperative agreement 1U54GM088558 (JT, JS), as well as NIEHS Center grant ES009089 (JS), the RAPIDD program of the Science and Technology Directorate, US Department of Homeland Security (JS), and the in-house influenza research program of the Fogarty International Center, NIH, funded by the Office of Global Affairs' International Influenza Unit, Office of the Secretary, US Department of Health and Human Services. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: JS discloses consulting for JT and Axon Advisors and partial ownership of SK Analytics.

Author Summary

An influenza pandemic virus emerged in North America in 2009. Although the virus spread worldwide within months of its emergence, the timing of peak pandemic activity varied by nearly a year across global populations. In fact, the most intense period of pandemic activity occurred earlier in temperate countries of the southern hemisphere (e.g., South Africa, Argentina and Australia) than in countries that reported the first infections (Mexico and the US). This indicates that the timing of peak pandemic activity did not directly align with the arrival of the virus in a location, suggesting the influence of social and environmental factors on pandemic influenza transmission. In this study we examined two large pandemic outbreaks ("waves") that occurred in summer and fall of 2009 in Mexico. The summer wave occurred in the tropical southeastern states of Mexico, whereas the larger fall wave occurred in the central and northern states. We find that the distinct pandemic waves were likely caused by complex interactions between regional differences in specific humidity, population level susceptibility, and school term cycles. Improving our understanding of the factors modulating regional transmission patterns will improve prediction of the spatial and temporal progression of future pandemics.

Introduction

There is strong evidence that seasonal influenza activity around the world is modulated by environmental variability $[\underline{1}]$. Temperate regions are characterized by annual winter epidemics $[\underline{2,3}]$ that may result from seasonal decreases in specific humidity and subsequent increases in virus survival and transmission $[\underline{4-6}]$. Seasonal influenza activity in the tropics is not as clearly phased $[\underline{7,8}]$, but tends to peak in seasons with high levels of specific humidity and rainfall $[\underline{1}]$. Yet, the extent to which these same factors affect the transmission of pandemic influenza remains largely unknown. The emergence of the novel A/H1N1pdm influenza virus in early 2009 and its subsequent global pandemic spread $[\underline{9}]$ provides a unique opportunity to examine these links.

The influenza A/H1N1pdm virus spread globally within weeks of the first report of a laboratory-confirmed case on April 19, 2009 in southern California [10]. The global dissemination of the virus was facilitated by global transportation networks from its putative source in North America [11], and by June 11, 2009 laboratory-confirmed infections had been identified in over 70 countries [12]. Although the virus had spread worldwide within months of its emergence, the country-to-country timing of peak pandemic activity varied by >40 weeks [13]. In fact, the most intense period of pandemic activity occurred earlier in temperate countries of the southern hemisphere (South Africa, Argentina and Australia) than in countries that reported the first A/H1N1pdm infections (Mexico and the US) [13]. These observations indicate that the timing of peak pandemic activity did not directly align with the arrival of the virus in a location, suggesting the influence of social and environmental factors on pandemic influenza transmission.

Several studies have examined the relative contribution of environmental drivers and school mixing on influenza transmission with conflicting results. School cycles have been shown to play a significant role in the transmission of seasonal influenza by modulating contact rates between children [14,15]. School closures have been linked to reductions in 2009 pandemic A/H1N1 transmission [16–18], while regional variability in school terms and weak child-mobility have been associated with the staggered occurrence of fall pandemic activity in US cities [17,19]. Specific humidity and prior immunity may have played a role in explaining a third

wave of pandemic activity during the winter of 2009–2010 in the southeastern US [20]. Similarly, the spatiotemporal patterning of the 2009 pandemic in Canada was associated with school terms and specific humidity [21,22]. In Chile, latitudinal variation in the timing of peak pandemic activity was associated with specific humidity but not with winter vacations, as pandemic activity was already subsiding when schools closed [23]. Altogether, the transmission impact of environmental forcing, population-scale interventions and school cycles has been broached but yet to be fully understood for pandemic influenza.

Here we fit Susceptible-Exposed-Infected-Recovered (SEIR) models to epidemiological data from Mexico using Markov Chain Monte Carlo (MCMC) approaches and investigate the importance of environmental forcing, school cycles, and interventions in generating the observed spatially-heterogeneous waves of 2009 pandemic influenza activity. Mexico is a compelling case study as it encompasses both temperate and tropical regions, interventions and school vacation periods were largely uniform across the country, and spatially-detailed epidemiological records are available [<u>16</u>].

Methods

Epidemiological and environmental data

We obtained daily laboratory-confirmed A/H1N1pdm influenza case data for April—December 2009 from a prospective epidemiological surveillance system that was established in response to the pandemic by the Mexican Institute for Social Security [16,24]. These data were previously used to document the regional patterns of pandemic activity in Mexico across 31 states and the Federal District (hereafter we refer to these as 32 "states") [16]. Circulation of the pandemic virus was intense during 2009 in Mexico, subsided by January 2010, and was followed by a period of 2-years of sporadic transmission [25]. Hence we capture the full extent of the first year of pandemic activity in this analysis.

A mandatory policy of school closure was strictly enforced for a 14-day period during April 27—May 11, 2009 across all states in Mexico [16]. In addition, our study period encompasses three school vacation periods, synchronous across Mexico, including a spring (April 5–18), summer (July 3—Aug 24) and winter break (beginning Dec 22; Fig 1). We retrieved and compiled daily specific humidity, precipitation and temperature data for the study period from the Global Land Data Assimilation System [26] for the most populous city in each state.

Pandemic patterns and descriptive statistical analysis

There were three spatially-heterogeneous pandemic waves in Mexico in 2009, including a spring wave from April 1—May 20, a summer wave from May 21—August 1, and a fall wave from August 2—December 31, as previously defined [16]. For each state, we calculated the cumulative case proportion of each wave (the number of cases during the wave relative to the total number of cases during the study period) and their association with average temperature, average humidity and cumulative precipitation conditions during each wave period using Pearson correlation. We then classified each location as dominated by a spring, summer or fall wave based on the week with the maximum number of cases.

Influenza transmission models

To further assess the dynamical effects of environmental variability and school cycles on pandemic influenza transmission, we developed a deterministic SEIR compartmental model at the state level. As a first step, we fit separate models for each state, and as a second step, we explore a meta-population framework allowing for coupling between states. The simplest formulation





Fig 1. Spatial and temporal structure of pandemic waves. Cumulative case proportions (i.e., the number of infections during a time period divided by the total number of infections during the study period) for the (*a*) spring, (*b*) summer and (*c*) fall waves by state. (*d*) Time series of daily case proportion for each state, and case proportions averaged across southeastern (red), and central and northern states (blue). Altogether, the plots indicate that the spring wave was relatively minor relative to the summer and fall waves; and that the summer and the fall waves were geographically distinct. Cumulative case proportions for the (*a*) spring, (*b*) summer and (*c*) faily case proportion for each state, and case proportions averaged across southeastern (red). Time series of daily case proportion for each state, and case proportions averaged across southeastern, and central and northern states. The plots indicate that the spring wave was relatively minor relative to the summer and fall waves; and that the spring wave was relatively minor relative to the summer and fall waves; and that the spring wave was relatively minor relative to the summer and fall waves; and that the spring wave was relatively minor relative to the summer and fall waves; and that the spring wave was relatively minor relative to the summer and fall waves; and that the spring wave was relatively minor relative to the summer and fall waves; and that the summer and the fall waves were geographically distinct.

doi:10.1371/journal.pcbi.1004337.g001

of our model (independence between states) is expressed by the following equations:

$$dS_i/dt = -\lambda_i S_i$$

 $dE_i/dt = -\lambda_i S_i - \theta^{-1} E_i$
 $dI_i/dt = \theta^{-1} E_i - \alpha^{-1} I_i$
 $dR_i/dt = \alpha^{-1} I_i$

where $S_{i, i} E_{i, i} I_{i, i} R_{i}$ are, respectively, the number of susceptible, exposed, infected, and recovered individuals in Mexican state *i*, θ is the mean latency period, α is the mean infectious period, and λ_{i} is the force of infection [27]. In the main analysis presented here, the mean latency period (θ) was fixed at 1.4 days based on past estimates [28], while the mean recovery rate (α) was fixed at 1.6 days, consistent with a mean serial interval of 3 days [29]. Further sensitivity analysis was performed in which the mean latency period (θ) was allowed to vary from 1–1.8

Parameter	Description	Refs.	Range	Model 1 (No spatial coupling, independent fit to each state)	Model 2 (meta- population model with coupling with edjacent states)	Model 3 (meta-population model with coupling with edjacent states and greater Mexico City hub)
W ₁	Value of specific humidity (g/kg) corresponding to minimum R ₀	[<u>1</u>]	6–14	х	x	x
W2	Minimum Ro	[<u>6]</u>	1–2	x	х	x
<i>W</i> 3	Added to w_2 to define R_0 for specific humidity = 0 g/kg	[6]	0–2	x	x	X
W4	Added to w_2 to define R_0 for specific humidity = 23 g/kg	[1]	0–2	x	x	X
Y1	Transmission efficiency during school vacation	[15]	0.55–0.90	x	x	x
Y2	Transmission efficiency during intervention and school closures	[<u>15</u> , <u>16</u>]	0.55–0.90	x	x	X
τ(0)	Rate of infected population at time = 0	NA	10 ⁻⁶ – 10 ⁻³	x	x	x
C _{adj}	Links force of infection between adjacent states	NA	0–1	_	x	x
C _{hub}	Links force of infection in hub states with all other states	NA	0–1	_	_	x
μ(0)	Pre-pandemic population susceptibility fraction	[32]	0.95	x	x	x
Θ	Latency period (days)	[28]	1.4	x	x	х
α	Mean infectious period (days)	[<u>28</u> , <u>29</u>]	1.6	x	x	x

Table 1. A description of the epidemiological parameters for each model. The references indicate the studies that informed the parameter estimates and ranges. The "x" indicates parameters that are included in the corresponding model.

doi:10.1371/journal.pcbi.1004337.t001

days, while the mean recovery rate (α) was allowed to vary from 0.6–2.6 days, consistent with a mean serial interval of 2.4–3.6 days [29].

We allowed transmission to vary spatially and temporally as a function of specific humidity, school terms and interventions. Following [6] and [20], we let the basic reproduction number, $R_{0,i}(t)$, vary with time as a function of daily state-specific specific humidity $q_i(t)$:

$$R_{0,i}(t) = f(q_i(t))$$

We chose this formulation for $R_0(t)$ because environmental forcing was the main factor under investigation in this study. The functional relationship between R_0 and specific humidity was defined by fitting a Piecewise Interpolating Polynomial (PCHIP) curve through three critical points (p_1, p_2, p_3) . PCHIP was used because it provided optimal control over the relationship by forcing the curve to pass through the specificed points without overshooting the defined minima or maxima. The critical points were defined by 4 free parameters (w_1, w_2, w_3, w_4) (Table 1). We specified that R_0 values at p_2 and p_3 to be greater than or equal to p_1 (Fig 2). Altogether, this specification of the points allowed for U-, J-, and L-shaped cruves, in addition to flat lines (see <u>S1 Text</u>).

Next, we let the effective reproduction number, $R_e(t)$, depend both on $R_0(t)$ the proportion of the susceptible population, $S_i(t)$, as well as school terms, v(t) and interventions, z(t). For school terms and interventions, we used step functions to represent changes in influenza



Fig 2. Specification of R_0 . A diagram showing the three critical points (p_1, p_2, p_3) used to define the relationship between specific humidity and R_0 . The points were allowed to vary across indicated ranges (dashed lines). The central point, p_1 , was allowed to vary along the x- and y-axes; whereas p_2 and p_3 only varied along the y-axis as 0 and 23 g/kg were the bounds of specific humidity.

doi:10.1371/journal.pcbi.1004337.g002

transmissibility, as follows:

$$v(t) = \begin{cases} 1, & if \text{ school term} \\ \\ \gamma_1, & if \text{ school vacation } 0.55 \le \gamma_1 \le 0.90 \end{cases}$$

$$z(t) = \begin{cases} 1, & \text{if no intervention} \\ \gamma_2, & \text{if intervention} \\ 0.55 \le \gamma_2 \le 0.90 \end{cases}$$

where γ_1 , and γ_2 are independent and bounded parameters (<u>Table 2</u>). Altogether, the effective reproduction number (R_e) in state *i*, follows:

$$R_{e,i}(t) = v(t)z(t)R_{0,i}(t)S_i(t)N^{-1}$$

Table 2. Summary of goodness-of-fit measures for select models. The table indicates the number of summer waves accurately predicted in the 6 states in which they were observed, the number of fall waves accurately predicted in the 32 states observed, and the AIC for each model. The parameters included in each model are described in Table 1.

Model	Summer Wave Predicted/ Observed	Fall Wave Predicted/ Observed	AIC
Model 1 (No spatial coupling, independent fit to each state)	6/6 (100%)	23/26 (88%)	-93805
Model 2 (Meta-population model with coupling with edjacent states)	2/6 (33%)	26/26 (100%)	-93267
Model 3 (Meta-population model with coupling with adjacent states and greater Mexico city hub)	3/6 (66%)	24/26 (92%)	-94165

doi:10.1371/journal.pcbi.1004337.t002

We also developed several flavors of meta-population models to explore travel effects. Based on past work indicating the importance of local diffusion on influenza transmission [19] we allow for neighboring states to affect the local force of infection. In the absence of mobility data from Mexico to calibrate a more detailed population model, we also allow for the greater Mexico City area to affect the risk of transmission in other areas, as the Mexican capital is a hub for both air and bus travel, the two dominant modes of transportation in the country. Our approach borrows from the concept of gravity models [30], whereby both large populations and nearby populations may affect the risk of dissemination to a new locale.

Specifically, we allow the force of infection in each state, λ_i , to be modified by the force of infection in neighboring states, λ_{adj} , and in two "hub" states (Mexico City and the Federal District), λ_{hub} . For each state, *i*, the λ_i at time *t* prior to mixing between states is defined as:

$$\lambda_i(t) = B_i(t)I_i(t)N^{-1}$$

where B_i is the transmission rate for state *i*. $B_i(t)$ and $R_{0,i}(t)$ are related by:

$$B_i(t) = R_{0,i}(t)\alpha^{-1}$$

When both travel effects are included, the force of infection in state *i* is modified by the force of infection in all adjacent states and two hub states as follows:

$$\hat{\lambda}_i(t) = \lambda_i(t) + c_{ ext{adj}} \sum_{j=1} \lambda_{ ext{adj}}(t) + c_{ ext{hub}} \sum_{k=1} \lambda_{ ext{hub}}(t)$$

where c_{adj} and c_{hub} are free parameters that are allowed to vary from 0–1.

Overall, we compare the fit of 3 increasingly complex spatial models: (a) models fitted independently to each state ($c_{hub} = c_{adj} = 0$), (b) meta-population models with nearest neighbors coupling ($c_{adj} > = 0$ and $c_{hub} = 0$) and (c) meta-population models with nearest neighbor coupling and hub centered around the capital ($c_{hub} > = 0$ and $c_{adj} > = 0$).

All models incorporate estimates of state population size from the National Council of Population, Mexico for 2009 [31]. Populations in each state were assumed to be fully mixed. The initial proportion of susceptibles, $\mu(0)$, was set to 95% across all states based on the age structure of the population in Mexico and estimates of worldwide prevalence of age-specific prepandemic cross-reactive antibody responses to A/H1N1pdm [32]. As a sensitivity analysis, the initial proportion of susceptible population was allowed to vary from 0.75–0.95. Further, in a separate model we allowed for spatial variation of $\mu(0)$ between southeastern states and central and northern states (see S1 Text). In all models, initial incidence in the simulations, $\tau(0)$, was uniform across states and allowed to vary from 1–1,000 per 100,000 people (Table 2).

The SEIR models were continuous and deterministic. The ordinary differential equations were solved numerically using Matlab version 8.2 (The Mathworks, Inc). We employed an adaptive Metropolis-Hastings algorithm to perform MCMC simulations and estimate parameter values [33]. We estimated model parameters (Table 2) by fitting the model-derived daily case proportion to empirical data for all states during the pandemic period. Using daily case proportion rather than case incidence allows standardization for potential reporting differences between states. We assumed uniform prior distributions for estimated parameters. We allowed the algorithm to run for 100,000 iterations following an initial burn-in of 250,000. The Geweke diagnostic method was employed to assess convergence of chains [34] with values close to 1 deemed satisfactory.

We compared the fit across models with the Akaike information criterion (AIC) using the observed and simulated daily time series across all states. We also assessed model fit by



Fig 3. Influenza cases and environmental variables by wave. Relationships between cumulative case proportion for spring, summer and fall waves and environmental variables. Strong relationships are observed between cumulative case proportions and environmental variables during the summer and fall waves; however, the relationship with precipitation is strongly influenced by outliers. Specific humidity has the strongest and most robust relationship for summer and fall waves. The relationships between the environment and the summer wave are opposite of those observed during the fall wave, likely due to spatial heterogeneity of population level susceptibility caused by the summer wave.

doi:10.1371/journal.pcbi.1004337.g003

examining whether peak pandemic incidence in each state occurred during the summer or fall in simulated data, and how this corresponded to the observed timing using the chi-square test.

Results

Multiple pandemic waves in Mexico: Descriptive analysis

There were three pandemic waves in Mexico during the 2009 pandemic (Fig 1). The spring wave was relatively minor and concentrated in the greater Mexico City area. The summer wave was predominant in southeastern states, and the fall wave was concentrated in central and northern states.

The cumulative case proportion during the summer wave was associated with higher mean specific humidity conditions (Pearson's correlation, r = 0.74, 95% CI: 0.63, 0.85). Mean temperature (r = 0.42, 95% CI: 0.20, 0.63) and rainfall (r = 0.59, 95% CI: 0.20, 0.78) were also significantly associated with summer cumulative case proportion, but were strongly influenced by outliers (Fig 3). The cumulative case proportion during the fall wave was negatively associated with specific humidity (r = -0.75, 95% CI: -0.86, -0.62), temperature (r = -0.48, 95% CI: -0.73, -0.37) and precipitation (r = -0.56, 95% CI: -0.70, 0.01). The significant association between precipitation and cumulative case proportion was primarily due to a single outlier (Fig 3).

Transmission models

We fit mechanistic transmission models to the progression of the pandemic in 32 states to assess the dynamical consequences of school cycles, social distancing interventions (that

PLOS | COMPUTATIONAL BIOLOGY



Fig 4. Model results. Time series of case proportion from April 1—December 31 for each state in Mexico during 2009 pandemic. The light blue/red lines correspond to 5000 simulations from random draws from the posterior distributions of each parameter. The dark blue/red lines corresponds to the best-fit simulation based on AIC. The black lines correspond to observed case proportions, and the shaded areas in background correspond to spring vacation, period of school closures and intervention measures, summer vacation, and winter vacation, respectively.

doi:10.1371/journal.pcbi.1004337.g004

include but are not limited to school closure, see Fig 1), and spatiotemporal variation in environmental drivers. Since the strongest statistical association between influenza activity and environmental forcing was seen with specific humidity in exploratory analyses (Fig 3), we allowed R_0 to vary flexibly as a function of specific humidity only. This aligns with laboratory and epidemiological evidence linking influenza activity with variations of specific humidity [1,5,6,20-23].

We considered a series of increasingly complex models that included different levels of spatial mixing. Simple transmission models including humidity, school and intervention terms, but no spatial coupling accurately identified the season of the largest pandemic outbreak in 29 of 32 states (P < 0.001; <u>Table 2</u>, model 1). The model also accurately described observed pandemic activity in Colima, where a relatively balanced proportion of cases occurred in summer and fall. However, the model overestimated early summer transmission in several central and northern states, in particular, Veracruz, Guerrero and Morelos where cases primarily occurred during the fall (Fig 4).

Comparison of AIC values across model structures indicated a significant improvement with the addition of a spatial mixing term allowing for interaction between adjacent states; however

the proportion of peak incidence correctly predicted with respect to season decreased (27/32; <u>Table 2</u>, model 2). Inclusion of a term representing connectivity with Mexico City did not improve model fit nor peak prediction accuracy (<u>Table 2</u>, model 3). The estimated relationship between R_0 and specific humidity, and the effect of school terms and interventions, were similar across all models. Since the model without spatial mixing best predicted pandemic activity, and estimates of the majority of other parameters did not change significantly with inclusion of spatial mixing terms, we focus on the model with no spatial mixing in subsequent sections.

Our MCMC estimation algorithm achieved high convergence with the exception of two parameters, w_1 and $\tau(0)$ which determine the value of specific humidity at which R_0 becomes minimum and the initial incidence of infection, respectively (Table 2 and S1 Text). For the best-fit model, the simulated infection attack rate averaged across all states was 28% and ranged from 26–34%.

Models where the mean latency period (θ), the mean recovery rate (α) and the initial level of susceptibility $\mu(0)$ were allowed to vary were not as robust as the primary models; however the models suggested a strong and consistent role for the effect of specific humidity and school terms on the pandemic waves (see <u>S1 Text</u>).

The impact of specific humidity and susceptibility on the occurrence of summer and fall waves

The estimated relationship between specific humidity and R_0 was J-shaped, with greatest R_0 at high levels of specific humidity and minimal R_0 at moderate levels of specific humidity (Fig 5; Table 3). Our estimated R_0 values ranged between 1.14 and 1.26, depending on specific humidity conditions.

Our model suggests that population level susceptibility was uniformly high and near initial levels at the beginning of the summer wave (Fig 5). In late May and June 2009, R_e was estimated to be greatest in the southeastern states (mean = 1.26) due to more humid conditions; whereas in the central and northern states R_e was slightly lower (mean = 1.21) due to moderate levels of specific humidity. The model implies that these regional differences in estimated R_e , driven by differences in specific humidity, were critical for the formation of the heterogeneous pandemic wave pattern and fomented greater incidence during summer in the more humid southeastern states.

During the summer vacation period, estimates of R_e decreased below 1 in nearly all states owing to school closure. At the beginning of the school term in late August 2009, estimated R_e averaged 1.11 for the majority of central and northern states, where susceptibility remained high (~85%). In contrast, susceptibility estimates in southeastern states were significantly lower (~75%), which reduced transmission rates (Fig 5E).

Effect of school vacation and social distancing on the occurrence of summer and fall waves

We estimate that school vacation decreased influenza transmission by 14% (95% CI: 10%, 19%) on average during the spring and summer breaks. During the 14-day period in May 2009 when stringent social distancing interventions were put in place, R_0 was reduced by 20% (95% CI: 11%, 40%); the large 95% credible interval likely has to do with a strong correlation between estimates of w_2 (minimum of R_0) and z (impact of interventions) in the model. Although we put constraints on the impact of intervention and school vacation periods based on past studies (Table 2 and [15,16]), relaxing these constraints did not change our estimates, attesting to the importance of reduced transmission during these periods. Indeed, simulations indicate that if



Fig 5. Variability of R_o and R_e during May-December 2009 as a function of changes in specific humidity, interventions, school cycles, and susceptibility, and the resulting impact on pandemic influenza activity, as predicted by the model. (A) The relationship between specific humidity and R_o . The gray lines are 500 samples generated from the posterior distributions, and the black line corresponds to the mean value of the posterior means. (B)

PLOS COMPUTATIONAL BIOLOGY

The time series of average specific humidity for central and northern states (solid line) and southeastern states (dashed line). (*C*) Time series of R_0 for best-fit parameter combination. The step features are related to spring break, the intervention period, summer break and winter break, respectively. (*D*) Time series of simulated R_e . (*E*) Time series of simulated population level susceptibility. (*F*) Time series of simulated case proportions. For *B*-*F*, shaded areas in background correspond to spring vacation, period of school closures and intervention measures, summer vacation, and winter vacation, respectively. The solid lines correspond to the central and northern states, and the dashed lines correspond to the southeastern states. The simulated values were generated using the mean value of the posterior means for each parameter.

doi:10.1371/journal.pcbi.1004337.g005

the summer vacations were eliminated, peak pandemic activity would have occurred in June-July 2009 for all states (see <u>S1 Text</u>).

Discussion

Our results indicate that the spatiotemporal structure of the 2009 influenza pandemic in Mexico can be understood by the interplay between specific humidity, school cycles, and susceptibility. Our results suggest that high specific humidity in the southeastern states allowed for relatively high transmission rates in late May-June 2009 and favored a substantial outbreak in these states (Fig 5); the pandemic wave in the southeastern states subsided during summer school vacations as R_e decreased. The results suggests that when school activities resumed in the fall, transmission increased due to increased contact rates and triggered outbreaks in some southeastern states, but these were relatively minor due to reduced levels of susceptibility following the summer wave. This is similar to the process that explains differences in the progression of the pandemic across regions in the US during the fall and winter of 2009–2010 [20].

Our model predictions indicate that central and northern states experienced more than 85% of their cases during the fall wave, consistent with observations. Our analysis suggests that the lack of substantial viral activity in this region in late May-June 2009 was due to slightly lower transmission rates associated with moderate levels of humidity, then compounded by school vacation, further reducing transmission rates. When summer vacation ended, the subsequent increase in contact rates made large outbreaks possible in this region. Our simulations suggest that a large summer wave would have occurred in the central and northern states if summer vacation had not reduced transmission (see <u>S1 Text</u>). This highlights that both environmental

Parameter	Model 1: no spatial coupling		Model 2: spatial coupling with adjacent states		Model 3: spatial coupling with adjacent states and greater Mexico City hub	
	Parameter Estimate (95% CI)	Geweke	95% CI	Geweke	95% CI	Geweke
W ₁	7.12 (6.02, 9.60)	0.18	7.81 (6.07, 10.44)	0.08	7.35 (6.01, 9.49)	0.04
W2	1.28 (1.24, 1.40)	0.90	1.59 (1.14, 1.76)	0.79	1.20 (1.01, 1.43)	0.66
W ₃	0.02 (0.00, 0.08)	0.83	0.10 (0.00, 0.40)	0.76	0.34 (0.10, 0.65)	0.94
W_4	0.16 (0.08, 0.26)	0.93	0.43 (0.21, 0.95)	0.82	0.84 (0.50, 1.12)	0.67
Y1	0.86 (0.80,0.89)	0.98	0.81 (0.76, 0.86)	0.97	0.80 (0.75, 0.85)	0.95
γ2	0.80 (0.60, 0.89)	0.98	0.79 (0.63, 0.89)	0.88	0.82 (0.65, 0.90)	0.99
τ(Ο)	1.14e-4 (1.05e-4, 1.36e-4)	0.57	1.24e-5 (4.19e-6, 2.06e-5)	0.88	9.49e-6 (3.10e-6, 1.67e-6)	0.30
C _{adj}	—	_	0.01 (0.00, 0.04)	0.68	0.03 (0.01, 0.05)	0.48
C _{hub}	—	—	—	—	0.01 (0.00, 0.02)	0.84
μ(0)	0.95	—	0.95	—	0.95	—
Θ	1.4	_	1.4	_	1.4	_
α	1.6	_	1.6	_	1.6	_

Table 3. Parameter estimates and corresponding confidence intervals, and the Geweke index are provided for models with and without mixing.

doi:10.1371/journal.pcbi.1004337.t003

variability and school cycles were critical for generating the distinct summer and fall wave patterns observed in Mexico.

We developed an SEIR model framework that allowed local levels of specific humidity to modulate transmission rates across a wide distribution of relationships, including a bimodal relationship indicating that influenza activity is enhanced by very low levels and very high levels of specific humidity [1]. Unlike previous studies that examined the effect of specific humidity (among other variables) on the progression of pandemic waves in the US [20], Canada [21,22], and Chile [23], it was necessary to use a bimodal relationship here because Mexico encompasses both temperate and tropical regions.

Previous studies have suggested that population mixing across states may have been a factor in the formation of the spring, summer and fall waves [35]. We developed a model that allowed for mixing between adjacent states and a "hub" region representing the greater capital area (Mexico and the Federal District). However, we found no evidence that connectivity with Mexico City, and hence a hierarchical pattern of spread, could explain the spatiotemporal structure of the pandemic waves. In contrast, connectivity with adjacent states was important, which is reminiscent of the slow diffusive pattern of 2009 pandemic in US cities [19]. This suggests that only local environmental conditions, social mixing and susceptibility patterns shaped the trajectory of the outbreak once the virus became established in the population.

We found no relationship between environmental conditions and the intensity of the spring wave of the 2009 pandemic in Mexico, which remained focused in central states. The spring wave may have only materialized in states highly connected to the geographic origin of the A/H1N1pdm virus prior to the initiation of intervention measures. We are unable to test this hypothesis further as the origins of the 2009 pandemic remain debated.

We estimated a number of disease parameters that compare favorably with independent information, reinforcing the validity of our modeling approach. In particular, our estimates of pandemic infection attack rates ranged from 26–34% for the cumulative period April-December 2009, which is commensurate with estimates derived from global serological surveys (both symptomatic and asymptomatic infections) [29]. Our estimated R_0 values, ranging between 1.14 and 1.26, align with estimates from previous studies in Mexico and elsewhere [9,21,28]. However, our estimated R_e values in the southeastern (1.2, summer) and in central-northern states (1.1, fall) are smaller than previous estimates for these regions, which may stem from different modeling assumptions [16]. Finally, we estimated that drastic social distancing interventions reduced transmission by 20% (95% CI: 14%, 40%) in Spring 2009 in Mexico while summer school vacations reduced transmission by 14% (95% CI: 10%, 19%), which is broadly consistent with previous estimates [16].

Our model is prone to a number of limitations. In the absence of prior information on baseline pre-pandemic immunity in Mexico, we considered prior immunity in baseline models to be spatially homogenous at 5%, informed by global age-specific serosurveys [29]. We did not explicitly incorporate the role of heterosubtypic immunity in mitigating the spread of the virus [22,36,37]. Indeed, previous modeling studies suggest that prior immunity from seasonal strains can inhibit the transmission of pandemic strains, thereby either delaying pandemic waves until immunity wanes or creating multi-wave patterns in a single population [38,39]. It is possible that prior immunity in Mexico varied across regions as a result of differential phasing of seasonal influenza across the southeastern and central-northern regions. Specifically, given that seasonal influenza activity typically occurs during winter in central and northern Mexico [40], the pandemic virus started disseminating directly following the circulation of seasonal influenza strains, potentially stimulating cross-subtype immunity and reducing transmission early during the pandemic. In contrast, 6–9 months may have passed since the most recent seasonal influenza activity in the southeastern region where seasonal influenza activity occurred in the summer [41], potentially resulting in populations with relatively lower prepandemic immunity levels. In sensitivity analyses, we explored the possibility that pre-pandemic susceptibility varied independently in northern and southern states. A model allowing for regional differences in pre-pandemic immunity performed better than a model without, supporting higher prior immunity in the northern and central states than in the southeastern states; however we have no specific biological information to support this model. Further, it should be noted the model with regional differences in pre-pandemic immunity benefitted from prior information on the spatial structure of the pandemic wave in different Mexican regions, data that were not provided to the other models tested. Overall, the epidemiological consequences of prior immunity to pandemic influenza remains heavily debated and a subject worthy of further experimental and modeling work [36-39,42].

Although our model accurately captured the overall spatial structure of the pandemic, correctly predicting the season of peak pandemic activity in 29 of 32 states, some details of the pandemic were missed. Specifically, in some southeastern states, simulated summer outbreaks lagged 2-4 weeks behind the observed outbreaks. Further, the model did not accurately describe the intense growth of fall outbreaks in states such as Sonora, Tlaxcala and Hidalgo (Fig 4). The lack of age-structure in our model and finer details of the relationship between influenza and environmental forcing could explain these differences. As descriptive analyses revealed high correlation between regional pandemic patterns and specific humidity, we selected specific humidity as the most likely driver of transmission in our mechanistic approach. In line with previous work [20], we did not further consider the putative effect of other environmental covariates, particularly temperature, due to high collinearity with specific humidity. Another possible source of error is that we used weather data corresponding to the largest population in each state. Specific humidity and other weather variables can vary significantly within states. By using weather data at the largest population center in each state we mitigated some of the effects of spatial climate variability, but it should be noted that these data may not accurately represent conditions for people living outside the population center. Further, we did not consider stochastic effects and assumed that these effects were limited due to our focus on the large populations of Mexican states in a pandemic period where incidence is high, following earlier work [43]. An alternative approach would be to do seed infected hosts at critical times prior to each wave as in [44], which would be important to consider for small populations. Another issue in developing meta-population models for Mexico is the lack of detailed mobility data. Although the addition of a hub centered around the greater Mexico city area did not seem to affect the dynamics of the 2009 pandemic, details of population mobility could be more important in inter-pandemic seasons [30]. Future work could concentrate on calibrating more detailed meta-population model to incidence and mobility data in Mexico.

Our results do not support a substantial increase in transmission at low levels of specific humidity, which has been observed in previous studies for both seasonal [1,5,6] and pandemic influenza [20–23]. However, pandemic activity was concentrated during April-October in Mexico when specific humidity levels were at moderate-to-high levels in a majority of states, making it difficult to assess the relationship between low levels of specific humidity and transmission. Another possibility is that—as discussed above—prior immunity in the spring may have been high in the northern (drier states) due to recent seasonal influenza transmission thereby inhibiting the spread of influenza during the period at the beginning of the pandemic when specific humidity was relatively low.

Finally, although we accounted for institutional intervention measures, we did not account for changes in personal behavior (e.g., hand washing, avoiding public spaces, masks) that may have varied across time and space. Indeed, behavior change may have contributed to the formation of multiple waves in the UK during the 1918–1919 influenza pandemic [45]. Limited

evidence indicates there were changes in travel behavior in Mexico in response to the pandemic [46]. Changes in social behavior, however, would not explain regional differences in pandemic patterns across Mexico unless these changes were regionally heterogeneous.

Overall, our results indicate that the occurrence of spatially-heterogeneous waves of the A/H1N1 pandemic virus in Mexico can be understood through consideration of local specific humidity conditions, susceptibility, and school-driven mixing patterns. The effect of humidity on pandemic influenza transmission in Mexico is consistent with a recent global model of seasonal influenza activity that stipulates a bimodal relationship between influenza and specific humidity, where transmission is favored by very high and very low levels [1]. Broadly, these findings suggest that a greater understanding of the mechanisms that drive inter-pandemic influenza epidemics may increase our capacity to predict the timing of major outbreaks associated with novel pandemic influenza viruses.

Supporting Information

S1 Text. Additional information regarding methods and results. (DOCX)

Author Contributions

Conceived and designed the experiments: JT CV JS GC. Performed the experiments: JT GC. Analyzed the data: JT GC. Wrote the paper: JT CV JS GC.

References

- Tamerius JD, Shaman J, Alonso WJ, Bloom-Feshbach K, Uejio CK, Comrie A, et al. Environmental Predictors of Seasonal Influenza Epidemics across Temperate and Tropical Climates. Riley S, editor. PLoS Pathog. Public Library of Science; 2013; 9: e1003194. doi: <u>10.1371/journal.ppat.1003194</u> PMID: <u>23505366</u>
- Finkelman BS, Viboud C, Koelle K, Ferrari MJ, Bharti N, Grenfell BT. Global Patterns in Seasonal Activity of Influenza A/H3N2, A/H1N1, and B from 1997 to 2005: Viral Coexistence and Latitudinal Gradients. Myer L, editor. PLoS ONE. Public Library of Science; 2007; 2. doi: <u>10.1371/journal.pone.0001296</u>
- Viboud C, Boëlle PY, Pakdaman K, Carrat F, Valleron AJ, Flahault A. Influenza epidemics in the United States, France, and Australia, 1972–1997. Emerging Infect Dis. 2004; 10: 32–39. doi: <u>10.3201/</u> <u>eid1001.020705</u> PMID: <u>15078594</u>
- Lowen AC, Mubareka S, Steel J, Palese P. Influenza Virus Transmission Is Dependent on Relative Humidity and Temperature. PLoS Pathog. Public Library of Science; 2007; 3: e151–1476. doi: <u>10.1371/journal.ppat.0030151</u>
- Shaman J, Kohn M. Absolute humidity modulates influenza survival, transmission, and seasonality. Proc Natl Acad Sci USA. 2009; 106: 3243–3248. doi: <u>10.1073/pnas.0806852106</u> PMID: <u>19204283</u>
- Shaman J, Pitzer VE, Viboud C, Grenfell BT, Lipsitch M. Absolute Humidity and the Seasonal Onset of Influenza in the Continental United States. PLoS Biology. 2010; 8: e1000316. doi: <u>10.1371/journal.</u> <u>pbio.1000316</u> PMID: <u>20186267</u>
- Moura FEA. Influenza in the tropics. Current Opinion in Infectious Diseases. 2010; 23: 415–420. doi: 10.1097/QCO.0b013e32833cc955 PMID: 20644472
- Viboud C, Alonso WJ, Simonsen L. Influenza in tropical regions. PLoS Med. 2006; 3: e89. doi: <u>10.1371/journal.pmed.0030089</u> PMID: <u>16509764</u>
- Fraser C, Donnelly CA, Cauchemez S, Hanage WP, van Kerkhove MD, Hollingsworth TD, et al. Pandemic potential of a strain of influenza A (H1N1): early findings. Science. 2009; 324: 1557–1561. doi: 10.1126/science.1176062 PMID: 19433588
- Centers for Disease Control and Prevention CDC. Swine Influenza A (H1N1) Infection in Two Children —Southern California, March—April 2009. In: MMWR Morbidity and mortality weekly report [Internet]. 24 Apr 2009 [cited 3 Oct 2013] pp. 400–402. Available: <u>http://www.cdc.gov/mmwr/preview/mmwrhtml/ mm5815a5.htm</u>

- Khan K, Arino J, Hu W, Raposo P, Sears J, Calderon F, et al. Spread of a Novel Influenza A (H1N1) Virus via Global Airline Transportation. N Engl J Med. 2009; 361: 212–214. doi: <u>10.1056/</u> NEJMc0904559 PMID: 19564630
- World Health Organization. New Influenza A (H1N1) virus: global epidemiological situation, June 2009. In: who.int [Internet]. 19 Jun 2009 [cited 3 Oct 2013] pp. 249–260. Available: <u>http://www.who.int/wer/2009/wer8425.pdf</u>
- Storms AD, van Kerkhove MD, Azziz-Baumgartner E, Lee W-K, Widdowson M-A, Ferguson NM, et al. Worldwide transmission and seasonal variation of pandemic influenza A(H1N1)2009 virus activity during the 2009–2010 pandemic. Influenza and Other Respiratory Viruses. 2013; 7: 1328–1335. doi: <u>10.</u> <u>1111/irv.12106</u> PMID: <u>23551904</u>
- Cauchemez S, Valleron A-J, Boëlle P-Y, Flahault A, Ferguson NM. Estimating the impact of school closure on influenza transmission from Sentinel data. Nature. Nature Publishing Group; 2008; 452: 750– 754. doi: 10.1038/nature06732 PMID: 18401408
- Hens N, Ayele GM, Goeyvaerts N, Aerts M, Mossong J, Edmunds JW, et al. Estimating the impact of school closure on social mixing behaviour and the transmission of close contact infections in eight European countries. BMC Infect Dis. 2009; 9: 187. doi: 10.1186/1471-2334-9-187 PMID: 19943919
- Chowell G, Echevarría-Zuno S, Viboud C, Simonsen L, Tamerius J, Miller MA, et al. Characterizing the Epidemiology of the 2009 Influenza A/H1N1 Pandemic in Mexico. PLoS Med. 2011; 8: e1000436. doi: <u>10.1371/journal.pmed.1000436</u> PMID: <u>21629683</u>
- Chao DL, Halloran ME, Longini IM. School opening dates predict pandemic influenza A(H1N1) outbreaks in the United States. J Infect Dis. 2010; 202: 877–880. doi: <u>10.1086/655810</u> PMID: <u>20704486</u>
- Huang KE, Lipsitch M, Shaman J, Goldstein E. The US 2009 A(H1N1) Influenza Epidemic. Epidemiology. 2014; 25: 203–206. doi: 10.1097/EDE.00000000000055 PMID: 24434751
- Gog JR, Ballesteros S, Viboud C, Simonsen L, Bjørnstad ON, Shaman J, et al. Spatial Transmission of 2009 Pandemic Influenza in the US. PLoS Computational Biology. Public Library of Science; 2014; 10: e1003635. doi: <u>10.1371/journal.pcbi.1003635</u> PMID: <u>24921923</u>
- Shaman J, Goldstein E, Lipsitch M. Absolute humidity and pandemic versus epidemic influenza. Am J Epidemiol. 2011; 173: 127–135. doi: <u>10.1093/aje/kwq347</u> PMID: <u>21081646</u>
- He D, Dushoff J, Eftimie R, Earn DJD. Patterns of spread of influenza A in Canada. Proc Biol Sci. 2013; 280: -20131174. doi: 10.1098/rspb.2013.1174 PMID: 24026815
- Earn DJD, He D, Loeb MB, Fonseca K, Lee BE, Dushoff J. Effects of School Closure on Incidence of Pandemic Influenza in Alberta, Canada. Annals of Internal Medicine. American College of Physicians; 2012; 156: 173–U32. doi: 10.7326/0003-4819-156-3-201202070-00005 PMID: 22312137
- Chowell G, Towers S, Viboud C, Fuentes R, Sotomayor V, Simonsen L, et al. The influence of climatic conditions on the transmission dynamics of the 2009 A/H1N1 influenza pandemic in Chile. BMC Infect Dis. 2012; 12: 298. doi: 10.1186/1471-2334-12-298 PMID: 23148597
- Echevarría-Zuno S, Mejía-Aranguré JM, Mar-Obeso AJ, Grajales-Muñiz C, Robles-Pérez E, González-León M, et al. Infection and death from influenza A H1N1 virus in Mexico: a retrospective analysis. The Lancet. 2009; 374: 2072–2079. doi: 10.1016/S0140-6736(09)61638-X
- Borja-Aburto VH, Chowell G, Viboud C, Simonsen L, Miller MA, Grajales-Muñiz C, et al. Epidemiological Characterization of a Fourth Wave of Pandemic A/H1N1 Influenza in Mexico, Winter 2011–2012: Age Shift and Severity. Archives of Medical Research. Elsevier Inc; 2012; 43: 563–570. doi: <u>10.1016/j.arcmed.2012.09.005</u> PMID: <u>23079035</u>
- Rodell M, Houser PR, Jambor UEA, Gottschalck J, Mitchell K, Meng CJ, et al. The global land data assimilation system. Bull Amer Meteor Soc. 2004; 85: 381–394.
- Diekmann O, Heesterbeek JAP. Mathematical Epidemiology of Infectious Diseases. John Wiley & Sons; 2000.
- Lessler J, Reich NG, Cummings DAT. Outbreak of 2009 Pandemic Influenza A (H1N1) at a New York City School. N Engl J Med. 2009; 361: 2628–2636. doi: <u>10.1056/NEJMoa0906089</u> PMID: 20042754
- 29. Boëlle P-Y, Ansart S, Cori A, Valleron A-J. Transmission parameters of A/H1N1 (2009) influenza virus pandemic: a review. Influenza and Other Respiratory Viruses. 2009; 5. doi: <u>10.1111/j.1600-0668.2009</u>. <u>00609.x</u>
- Viboud C, Bjørnstad ON, Smith DL, Simonsen L, Miller MA, Grenfell BT. Synchrony, waves, and spatial hierarchies in the spread of influenza. Science. American Association for the Advancement of Science; 2006; 312: 447–451. doi: 10.1126/science.1125237 PMID: 16574822
- National Council of Population, Mexico. 2015. Database: Estimates [Internet]. Available from: <u>http://www.conapo.gob.mx/es/CONAPO/Proyecciones_Datos</u>.
- van Kerkhove MD, Hirve S, Koukounari A, Mounts AW, H1N1pdm serology working group. Estimating age-specific cumulative incidence for the 2009 influenza pandemic: a meta-analysis of A(H1N1)pdm09

serological studies from 19 countries. Influenza and Other Respiratory Viruses. 2013; 7: 872–886. doi: 10.1111/irv.12074 PMID: 23331969

- Haario H, Laine M, Mira A, Saksman E. DRAM: efficient adaptive MCMC. Statistics and Computing. Springer; 2006; 16: 339–354.
- Geweke J. Evaluating the Accuracy of Sampling-based Approaches to the Calculation of Posterior Moments. 1991.
- **35.** Castillo-Chavez C, Cruz-Aponte M, Herrera-Valdez M. Multiple outbreaks for the same pandemic: Local transportation and social distancing explain the different "waves" of A-H1N1pdm cases observed in México during 2009. MBE. 2011; 8: 21–48. doi: 10.3934/mbe.2011.8.21 PMID: 21361398
- Ferguson NM, Galvani AP, Bush RM. Ecological and immunological determinants of influenza evolution. 2003; 422: 428–433. doi: <u>10.1038/nature01509</u> PMID: <u>12660783</u>
- Sridhar S, Begom S, Bermingham A, Hoschler K, Adamson W, Carman W, et al. Cellular immune correlates of protection against symptomatic pandemic influenza. NATURE MEDICINE. 2013. doi: <u>10.1038/</u> <u>nm.3350</u>
- Mathews JD, McCaw CT, McVernon J, McBryde ES, McCaw JM. A Biological Model for Influenza Transmission: Pandemic Planning Implications of Asymptomatic Infection and Immunity. Monk N, editor. PLoS ONE. 2007; 2: e1220. doi: <u>10.1371/journal.pone.0001220</u> PMID: <u>18043733</u>
- Mathews JD, McBryde ES, McVernon J, Pallaghy PK, McCaw JM. Prior immunity helps to explain wave-like behaviour of pandemic influenza in 1918–9. BMC Infect Dis. BioMed Central Ltd; 2010; 10: 128. doi: 10.1186/1471-2334-10-128 PMID: 20497585
- Chowell G, Bertozzi SM, Colchero MA, Lopez-Gatell H, Alpuche-Aranda C, Hernandez M, et al. Severe respiratory disease concurrent with the circulation of H1N1 influenza. New England Journal of Medicine. 2009; 361: 674–679. doi: 10.1056/NEJMoa0904023 PMID: 19564633
- Ayora-Talavera G, Góngora-Biachi RA, López-Martínez I, Moguel-Rodríguez W, Pérez-Carrillo H, Vázquez-Zapata V, et al. Detection of human influenza virus in Yucatan, Mexico. Rev Invest Clin. 2002; 54: 410–414. PMID: 12587415
- Mathews JD, Chesson JM, McCaw JM, McVernon J. Understanding influenza transmission, immunity and pandemic threats. Influenza and Other Respiratory Viruses. Blackwell Publishing Ltd; 2009; 3: 143–149. doi: 10.1111/j.1750-2659.2009.00089.x PMID: 19627371
- 43. Dorigatti I, Cauchemez S, Ferguson NM. Increased transmissibility explains the third wave of infection by the 2009 H1N1 pandemic virus in England. Proc Natl Acad Sci USA. 2013; 110: 13422–13427. doi: 10.1073/pnas.1303117110 PMID: 23882078
- Bolton KJ, McCaw JM, McVernon J, Mathews JD. The influence of changing host immunity on 1918– 19 pandemic dynamics. Epidemics. Elsevier B.V; 2014; 8: 18–27. doi: <u>10.1016/j.epidem.2014.07.004</u> PMID: <u>25240900</u>
- Fenichel EP, Kuminoff NV, Chowell G. Skip the Trip: Air Travelers' Behavioral Responses to Pandemic Influenza. PLoS ONE. Public Library of Science; 2013; 8: e58249. doi: <u>10.1371/journal.pone.0058249</u> PMID: 23526970
- 46. He D, Dushoff J, Day T, Ma J, Earn DJD. Inferring the causes of the three waves of the 1918 influenza pandemic in England and Wales. Proc Biol Sci. The Royal Society; 2013; 280: –20131345. doi: <u>10.</u> <u>1098/rspb.2013.1345</u> PMID: <u>23843396</u>